My (Yury Neretin) participations on international mathematical conferences related to the project P19064 (Hompgeneous spaces

Updated 10.06.10

Invited lectures

1. Stein-Sahi kernels.

Workshop “Complex analysis, Operator Theory, and applications to Mathematical Physics”, ESI, Vienna; October, 2006 2. Affine isometric actions and polymorphisms.

Workshop “Amenability beyond groups”, ESI, Vienna; March, 2007 3. Space of holomorphic functionals on univalent functions as a model of a boson Fock space Conference “New trends in complex and harmonic analysis”;, Bergen, Norway May 2007 4. Space of univalent functions as infinite dimensional analog of Cartan domain. Program “Representation theory, complex analysis and integral geometry”, Max- Planck-Institute, Bonn, June, 2007 5-6. Berezin kernels.

Inverse limits of classical groups.

Russian–Japanese workshop“Quantization and representation theory” Tambov, Russia; September, 2007 7-8. Gaussian integral operators. Virasoro algebra and univalent functions.

Workshop ”Virasoro algebra”, Bergen University, October, 2007 9. Inverse limits of classical groups.

Meeting ”Representation theory”, Oberwolfach, October, 2007 10. On natural kernels on symmetric spaces. “Seminar Sophus Lie”, Vienna, November 2007.

11. Gauss-Berezin operators and spinor representation of the supergroup OSp(2p|2q). Twente conference in Lie groups, Enscheide, Holland, December, 2007 12. Harmonic analysis of Berezin kernels.

Japanese–Russian workshop “Quantization and representation theory”, Fukuoka, Japan, February, 2008 13. Hilbert spaces of holomorphic functions and Berezin kernels. Workshop “Complex analysis and mathematical physics”, Sandane, Norway, June, 2008 14. Action of braid group on point configurations in 3-dimensional sphere., Tambov, April 2009 15. ”Point configurations on 3-dimensional sphere and outer automorphisms of free groups” Workshop on combinatorics, physics and mathematics, Moscow, June 2009

16. Double coset multiplication on infinite-dimensional classical groups, ”Classical groups seen from infinity.”, Moscow, August 2009